Author's response to reviews

Title: Primary lung tumor visualised by transthoracic echocardiography

Authors:

Magnus Dencker (magnus.dencker@skane.se)
Carin Cronberg (carin.cronberg@skane.se)
Sabine Damm (sabine.damm@skane.se)
Sven Valind (sven.valind@skane.se)
Monica Wadbo (monica.wadbo@skane.se)

Version: 2 Date: 15 December 2008

Author's response to reviews:

Dear Editor-in-Chief, Prof Picano

Thank you for considering manuscript titled “Primary lung tumor visualised by transthoracic echocardiography” for publication in the Cardiovascular Ultrasound. We have read the comments by the reviewer and made changes accordingly. One movie from the echocardiography examination has been added. The images from the echocardiography examination were however very poor. The tumour was best visualized from apical four-chamber view, other projections did not add any relevant information. We have also made an attempt to improve the language in the manuscript. We have also changed the references in the revised manuscript so that they conform to the journal style.

Yours truly
Magnus Dencker
Dept of Clinical Physiology and Nuclear Medicine
University Hospital MAS
205 02 Malmö, Sweden